Author's response to reviews

Title: Impact of singlehood during pregnancy on dietary intake and birth outcomes- a study in the Norwegian Mother and Child Cohort Study

Authors:

Jorunn Farbu (jorunn.farbu@gmail.com)
Margaretha Haugen (margaretha.haugen@fhi.no)
Helle Margrete Meltzer (helle.margrete.meltzer@fhi.no)
Anne Lise Brantsaeter (anne.lise.brantsaeter@fhi.no)

Version: 3
Date: 11 November 2014

Author's response to reviews: see over
Dear editor, BMC Pregnancy and Childbirth.

We are delighted that our manuscript MS 3554453141414721 “Impact of singlehood during pregnancy on dietary intake and birth outcomes- a study in the Norwegian Mother and Child Cohort Study” has been accepted for publication in BMC Pregnancy and Childbirth.

The editorial revisions required have been addressed as follows.

Editorial Comment:
1. Author's Contribution: Please include all authors in the author's contribution section.
   Answer: An additional sentence has been added ensuring that the initials of all four authors are mentioned. The author’s contribution section now reads:
   “All authors planned the study. MH calculated all food and nutrient intakes. JF conducted the statistical analyses and drafted the paper. ALB and MH assisted with statistical analyses. All authors contributed to the interpretation of the results. HMM contributed with writing of the paper and all authors reviewed the paper and approved the final manuscript.”

2. Figures in the manuscript: Please remove the figure from the main manuscript and upload it as a figure file.
   Answer: This has been done.

We trust that these revisions have been satisfactorily addressed and thank you for a swift and constructive review process.

Yours sincerely, on behalf of the authors,

Jorunn Farbu,
Norwegian Institute of Public Health, Division of Environmental Medicine